Author's response to reviews

Title: Obligations of low income countries in ensuring equity in global health financing

Authors:

John Barugahare (john.barugahare@uib.no)
Reidar K Lie (reidar.lie@fil.uib.no)

Version: 2 Date: 14 May 2015

Author's response to reviews:

Cover page with names and address of authors has been added
Acknowledgment section has been added
Format is as far as we can see in BMC style